Author's response to reviews

Title: In vitro activities of natural products against oral Candida isolates from denture wearers

Authors:

Cristina Marcos-Arias (cristina.marcos@ehu.es)
Elena Eraso (elena.eraso@ehu.es)
Lucila Madariaga (lucila.madariaga@ehu.es)
Guillermo Quindos (guillermo.quindos@ehu.es)

Version: 8 Date: 24 November 2011

Author's response to reviews:

To the BMC Series Editorial Production Team

Dear Sirs and Madams,

We made three small typographical changes. There are the following:

Page 3: We have corrected 'derivates' by 'derivatives': The aim of the current study was to investigate the in vitro activity of eight terpenic derivatives...

Page 4: We have changed the location of the Figure 1 reference: ... from Artemisia annua, and tyrosol (4-(2-hydroxyethyl)phenol) from Olea europaea (Figure 1). The compounds were...

Page 11: We have changed 'because' by 'due to': ... with those reported by other authors due to the different in vitro susceptibility tests used and the potential variation of susceptibilities among the isolates...

Looking forward to hearing from you

Guillermo